Author's response to reviews

Title: Supply side dimensions and dynamics of integrating HIV testing and counselling into routine antenatal care: A facility assessment from Morogoro Region, Tanzania

Authors:

Selena J An (sjan@jhu.edu)
Asha S George (ageorg22@jhu.edu)
Amnesty E LeFevre (aelefevre@gmail.com)
Rose Mpembeni (rcmpembeni@gmail.com)
Idda Mosha (ihmosha@yahoo.co.uk)
Diwakar Mohan (dmohan3@jhmi.edu)
Ann Yang (annyang17@gmail.com)
Joy Chebet (jchebet1@jhu.edu)
Chrisostom Lipingu (chrisostom.lipingu@jhpiego.org)
Abdullah H Baqui (abaqui@jhu.edu)
Japhet Killewo (jkillewo@yahoo.co.uk)
Peter J Winch (pwinch@jhu.edu)
Charles Kilewo (kilewo1950@gmail.com)

Version: 7 Date: 22 September 2015

Author's response to reviews: see over
September 21, 2015

Ms. Armee Valencia
Journal Editorial Office
BioMed Central

Dear Ms. Valencia,

Thank you for providing us with additional comments from one of the peer reviewers for our manuscript, “Supply side dimensions and dynamics of integrating HIV testing and counselling into routine antenatal care: A facility assessment from Morogoro Region, Tanzania.” The peer reviewer has provided us with very helpful comments, and we have made the corrections and provided additional clarification wherever applicable. Please find below a summary of reviewer comments along with our revisions and responses.

Comments from Reviewer 2

<table>
<thead>
<tr>
<th>#</th>
<th>Comments</th>
<th>Revisions and Responses</th>
</tr>
</thead>
</table>
| 1 | * pg 9 - replace the use of single-digit numbers with words per normal publication standards  
* pg 18 - Para 2 and para 3 you slip into present tense; fix this to make it past tense  
* pg 19 - it is unclear how one could observe 20 or fewer out of 203? is this a typo? | Thank you for pointing these out; we have made corrections and provided further clarification. |

Comments from Journal Editors

<table>
<thead>
<tr>
<th>#</th>
<th>Comments</th>
<th>Revisions and Responses</th>
</tr>
</thead>
</table>
| 1 | Please also ensure that your revised manuscript conforms to the journal style  
(http://www.biomedcentral.com/info/ifora/medicine_journals). It is important that your files are correctly formatted. | Thank you for the comment; we have edited the manuscript to ensure that it conforms to the journal’s style and formatting. |

Please see the updated manuscript and figures in the uploaded files. If you have any further concerns or questions, please do not hesitate to contact me at sjan@jhu.edu.

Thank you for your time and consideration. I look forward to hearing from you soon.

Best regards,

Selena An